Author's response to reviews

**Title:** Home Exposure to Arabian Incense (Bakhour) and Asthma Symptoms in Children: A Community Survey in Two Regions in Oman

**Authors:**

Omar A Al-Rawas (orawas@squ.edu.om)
Abdullah A Al-Maniri (almaniri@gmail.com)
Bazdawi M Al-Riyami (bazdawi@squ.edu.om)

**Version:** 6 **Date:** 12 May 2009

**Author's response to reviews:**

all the requested final changes were made. There is no Table 5, the text was referring to Table 4. this was corrected. the entire manuscript was also carefully reviewed fo any typo errors.